Author's response to reviews

Title: Solid variant of aneurysmal bone cyst of the heel; a case report

Authors:

   Joanna Lekka (Jlekka@yahoo.gr)
   Theofani Gavressea (fangav@261.gr)
   Gabriella Stanc-Giannakopoulos (gabidr1999@yahoo.com)
   Nikolaos Demertzis (NDemertzis@yahoo.gr)

Version: 8 Date: 27 November 2010

Author's response to reviews: see over
Author's response to reviews

Solid variant of aneurysmal bone cyst of the heel: a case report

J. Lekka (Jlekka@yahoo.gr)
Th. Gavresea (fangav@yahoo.gr)
G.Stanc-Giannakopoulos (gabidr1999@yahoo.com)
N. Demertzis (Ndemertzis@yahoo.gr)

Author response to review: see over

Restructured abstract:
Editor's report

1) 'The authors do need to revise the manuscript so that the conclusion is not so long. My feeling is that much of the text in the Conclusion section better fits into the Discussion section'

Changes made as indicated from the editors and we moved some of the Conclusion text into the Discussion section.

2) Aside from the editor's comment, we would like you to also address the following formatting changes:

- Please be sure that the title page should only include the title, authors, author affiliations and corresponding author details. The corresponding author should also be indicated. No other information should be included on this page. 'Abstract' section should start on page 2 of the manuscript

Changes made as you indicate.

3). Please be sure to include the ethnicity of the patient in the 'Abstract' section.

Changes made as you indicate

- Please provide a section heading/header in the 'Competing Interest' and 'Consent' section of your manuscript

Changes made as you indicate

All changes are highlighted with red colour.

Version 8: 27 November 2010